Author’s response to reviews

Title: A Prospective, Randomized Comparison of Ultrasonographic Visualization of Proximal Intercostal Block vs Paravertebral Block

Authors:

Kamen Vlassakov (kvlassakov@bwh.harvard.edu)
Avery Vafai (AVAFAI@PARTNERS.ORG)
David Ende (endeda@gmail.com)
Megan Patton (mpatton2@bwh.harvard.edu)
Sonia Kapoor (soniamk74@yahoo.com)
Atif Chowdhury (atifnchowdhury@gmail.com)
Alvaro Macias (macias@bwh.harvard.edu)
Jose Zeballos (jzeballos@bwh.harvard.edu)
David Janfaza (djanfaza@bwh.harvard.edu)
Sujatha Pentakota (spentakota@bwh.harvard.edu)
Kristin Schreiber (klschreiber@bwh.harvard.edu)

Version: 2 Date: 31 Dec 2019

Author’s response to reviews:

Editor Comments:

1. Trial Registration Information

Thank you for registering the study in a trial registry. Please ensure that the number and date of registration is included in the abstract. The last section of the abstract should be Trial Registration: listing the trial registry and the unique identifying number, e.g. Trial registration: Current Controlled Trials ISRCTN73824458, as well as the date of registration. Please note that there should be no space between the letters and numbers of the trial registration number. If registration took place after the first participant was enrolled, please state also “Retrospectively registered” at the end of this section
Response 1: This is included at the end of the abstract as NCT02911168

2. Consent for publication

In the Consent for publication statement of the Declarations, please clarify whether the consent to publish the images was written or verbal.

Response 2: This was included in the written consent form - we have clarified this in the revised manuscript in the declarations section.

3. All authors read and approved

Please include a statement in the Authors’ contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Response 3: We have added this in the revised manuscript in the authors’ contributions section.

*At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strike-throughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response 4: We have done so.

Regarding the reference #5, which review showed are unverified - if it is helpful, the PMID is 15248630 on pubmed.